Author’s response to reviews

Title: Fifty Years of Fat: News Coverage of Trends that Predate Obesity Rates

Authors:

Brennan Davis (bdavis39@calpoly.edu)
Brian Brian Wansink (fblsubmissions@cornell.edu)

Version: 2  Date: 30 December 2014

Author’s response to reviews:

Dear Dr. Latkin,

In addition to the changes we made as detailed in our letter below, we have further modified the Abstract to be more consistent with your guidelines, changing Objective to Background, and adding more context.